Author's response to reviews

Title: Itch and skin rash from chocolate during fluoxetine and sertraline treatment: Case report

Authors:
Jonas Cederberg (jonas.cederberg@medsci.uu.se)
Stefan Knight (stefan@xray.bmc.uu.se)
Svante Svenson (svante.svenson@lul.se)
Hakan Melhus (hakan.melhus@medsci.uu.se)

Version: 4 Date: 29 October 2004

Author's response to reviews:

1. Formatting changes according to the list provided by the Assistant Editor.
2. Background: "rash" is altered to "exanthema" as suggested by the referee and "urticaria", "pruritus" and "erythema multiforme" have been added.
3. Case presentation: "He had not previously..." is altered to "The patient did not have any history of allergy or dermatological diseases"
4. Abstract "...speculatively caused by a well-developed serotonergic system" removed.
5. Background: "...SSRI-medication have also been reported..." altered to ", "...SSRI-medication has also been used..."
6. Background: Last sentence altered to " The case is presented and we provide a putative biological rationale for the described phenomenon."
7. Case presentation: Sentence starting with "He now realised..." is altered to ". "He now remembered that he had had a chocolate mousse dessert before the first episode"
8. Case presentation: Sentence starting with "However, the results" is altered to " However, attempts to treat pruritus using 5-HT3-receptor-antagonists have not given clear-cut results [6,7,8]"
9. Case presentation: Sentence starting with "Chocolate contains..." is altered to " Chocolate contains serotonin, at concentrations which depends on the type of chocolate [14]"